Author's response to reviews

Title: The influence of Methylphenidate on the power spectrum of ADHD children - a MEG study

Authors:

Christian Wienbruch (Christian.Wienbruch@uni-konstanz.de)
Isabella Paul (Isabella.Paul@uni-konstanz.de)
Susanne Bauer (sudebauer@surfeu.de)
Hermann Kivelitz (Kivelitz-Ettlingen@t-online.de)

Version: 4 Date: 23 July 2005

Author's response to reviews: see over
Dear Dr. Hodkinson,

Thank you for accepting our manuscript “The influence of Methylphenidate on the power spectrum of ADHD children - an MEG study” for publication in BMC Psychiatry. Also we like to thank the reviewers for their most helpful and valuable comments regarding the manuscript. Please find enclosed our list of changes.

With kindest regards,

Christian Wienbruch
Changes made:

1. Affiliations have been changed and Email address of one of us has been modified.
2. Blank page has been removed
3. All references in the abstract were removed and the corresponding passage of the abstract has been changed slightly.
4. Page breaks have been removed
5. References (APA, 1994), (NIH consensus statement …), (Mental Health: a report of …) and (United Nations International Narcotics …) were added to the reference list and citation has been changed according to your suggestions. Therefore all citation numbers were changed in the text as well as in the reference list.
6. “Methods ADHD” has been changed to “Methods”
7. Webside adress www.wma.net was added to the reference list and is cited in the text now.
8. Section headings have been changed to sentence case
9. Figure placement text has been removed
10. All figures were renumbered and references in the text were changed accordingly.
11. “Authors’ contribution” has been slightly modified
12. “Acknowledgements” has been slightly modified.